Author's response to reviews

Title: The effects of adding epinephrine to ropivacaine for popliteal nerve block on the duration of postoperative analgesia: a randomized controlled trial

Authors:

Karin PW Schoenmakers (k.schoenmakers@maartenskliniek.nl)
Maaike GE Fenten (m.fenten@maartenskliniek.nl)
Jan Willem Louwerens (j.louwerens@maartenskliniek.nl)
Gert Jan Scheffer (gertjan.scheffer@radboudumc.nl)
Rudolf Stienstra (r.stienstra@maartenskliniek.nl)

Version: 2 Date: 31 May 2015

Author's response to reviews:

Thank you for the evaluation of our revised manuscript and for the constructive comments made by the reviewers.

Enclosed, the comments made by the reviewers are repeated verbatim in parentheses, our reply in blue. In the revision, all changes are delineated by bolded text. We believe that we have addressed all issues raised by the reviewers adequately and hope you will agree.

Thank you once again for your trouble; we look forward to your reply.